Author’s response to reviews

Title: The prevalence of angina symptoms and association with cardiovascular risk factors, among rural, urban and rural to urban migrant populations in Peru

Authors:

M. Justin S Zaman (drijustinzaman@hotmail.com)
Christian Loret de Mola (christian.loretdemola@mhperu.org)
Robert H Gilman (gilmanbob@gmail.com)
Liam Smeeth (Liam.Smeeth@lshtm.ac.uk)
J. Jaime Miranda (mirandajj@gmail.com)

Version: 5 Date: 1 October 2010

Author’s response to reviews:

Dear editors. This version has tiny and minor proof-reading changes. Thank you.
Dr MJS Zaman and co-authors